Author's response to reviews

Title: Reliability of capturing foot parameters using digital scanning and the neutral suspension casting technique

Authors:

Matthew Carroll (matthew.carroll@aut.ac.nz)
Mary-Ellen Annabell (maryellen.annabell@gmail.com.au)
Keith Rome (krome@aut.ac.nz)

Version: 4 Date: 28 February 2011

Author's response to reviews:

Major revisions

1. Email addresses: please make sure that the authors email addresses are identical in both the manuscript and submissions system.

The email address order had been amended

Minor revisions

2. Please re-order the author list in the manuscript so it matches the submissions system.

Order amended

3. Tracking: Please remove all tracking from the manuscript.

Tracking removed

4. Please remove the additional files section as you do not have any additional files.

Additional files section removed